EXERCISES 557

Exercises

Exercise 27.1 Approximate period tripling renormalization:

Implement an approximate renormalization for period-tripling sequence of figure 27.6, using the the approximate period-doubling renormalization scheme of example 26.2, applied to the complex polynomial

z_{n+1} = λ − z^2_n .

The idea is to drop from f_λ ∘ f_λ ∘ f_λ(z) all terms higher than z^2 in each period-tripling step. Evaluate numerically the complex coordinate and parameter rescaling universal constants α_{1/3} and δ_{1/3}.

(O.B. Isaeva and S.P. Kuznetsov)

ChaosBook.org/version11.8, Sep 7 2006 exerComplex - 15sep2006